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KINGSTON ASH RECOVERY PROJECT

WEEKLY REPORT
May 10-16, 2010

HIGHLIGHTS

e Cumulatively, just over 3,321,500 cubic yards of ash have been removed from the river and east of Dike 2.
e To date, 233 unit trains have been loaded and just over 2,214,400 tons of ash have been transported to Uniontown,
Alabama.

SAFETY

Program Summary
o 12 site specific HSE orientations were conducted for the week — 1,541 to date.
o 21 Safety Observation Reports submitted for the week: 1,144 to date; 1,081have been closed.

Recordable Incident:
e May 13, 2010 — Sevenson employee bitten by an insect, possibly a spider. The First Aid E)aramedic sent him to a
clinic where medication was prescribed. Employee returned to work on Monday, May 17".

INFRASTRUCTURE/ASH MANAGEMENT

e Efforts for site maintenance, dust control, and HAZWOPER control continued.

e Ash generated from time-critical activities was removed from the valley area early in the week and taken to be loaded.

e Concurrence was received on completion of the east embayment. Minor cleanup activities such as moving the
exclusion zone fencing and piping were completed.

e Held a pre-bid meeting with contractors on construction of the Swan Pond Circle Road underpass bridge and
abutments.

e Mayse Construction continued installation of the offsite utilities and construction is 65% complete.

ASH DREDGING & PROCESSING

o Atotal of 53,479 cy of ash were removed from the river this week by Sevenson. All dredges were operational.

e Mechanical dredging north of the site continued this week by Sevenson and Aquarius Marine. Ash is being taken to
the lateral expansion area for storage.

e The 10-inch TVA dredge in the ash pond was redirected to the lateral expansion area for discharge. Ash is being
recovered by dipping in a rim ditch fashion in the area. The ash dries on the platform and is then taken to the valley
area for storage.

e  Work continued dipping from the rim ditch and sluice trench. Lime treatment at the sluice trench continued this week
using Mt. Carmel. Treated ash was hauled directly to the trains for disposal.

o Efforts continued in the ballfield to move ash to the tracks and to dry the windrows. Much of the ballfield has been
emptied.

ASH DISPOSITION

e Loaded and transported 7 unit trains again this week.
e Have loaded and transported 233 unit trains for a total of approximately 2,214,400 tons.

e Continued inspection, maintenance, and repair of the rail in the rail yard. The repair work includes replacing rail and
ties, installing rail anchors, checking and re-gauging rail, and identifying additional ties for replacement.

e Continued monitoring moisture content of the ash and placing polymer in rail cars prior to shipping to Alabama.

CENOSPHERE RECOVERY

e Boat crews responded to outreach calls along the Tennessee and Clinch Rivers. There are patches of moderate
cenospheres resulting from the high water flushing cenospheres from the shore.

e Boat crews worked the past weekend cleaning shorelines along the Tennessee River.

e Approximately 12 outreach calls have been generated since the high waters. Over half have been resolved. The
outreach center continues to receive about one call per day.
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¢ Ash Pond conditions continued to improve. Crews continue to focus on keeping the weir areas clear.

SKIMMER WALL

e Agquarius Marine LLC, Skimmer Wall Debris Removal contractor, continued rigging the first caisson for removal from
the Intake Channel. All rigging complete and trial lift completed. Full extraction will be attempted May 17",
o Skimmer Wall Reconstruction contractor notified with award being made May 17",

DIKE REINFORCEMENT

o Dike C Buttress work continued. Civil Projects has crossed the dredge lines and is progressing south.

e Bulttress construction is about 27% complete.

e Continued to review Segment B drawings. A meeting is scheduled for May 18" to discuss construction schedule and
methods in the Intake Channel with the plant.

e Continued to monitor the dikes at Kingston on a daily basis.

ROUTINE MONITORING

Surface Water Sampling

e Routine surface water samples were collected at 10 fixed locations on the Emory, Clinch, and Tennessee Rivers on
Monday, May 10", and Friday, May 17". EPA collected split samples on May 17"

e Surface water analytical results are compared to the Tennessee Drinking Water Standards (TDWS) and the Ambient
Water Quality Criteria for Fish and Aquatic Life (F&AL) to establish a reference point for evaluating the potential
significance of the data. Although the most impacted section of the Emory River has been closed to public use, the
Emory, Clinch, and Tennessee Rivers are classified for use for Recreation and as a source for Domestic Water
Supplies. However, closure of the most impacted reach of the Emory River during the ash recovery operation must
be considered in these comparisons.

o Noteworthy results are reported below for Fixed River Location samples collected between 03/01/2010 and
04/19/2010. Comparisons to the F&AL are performed for dissolved concentrations only.

o Onthe Tennessee River, the arsenic was detected in 2 of 16 samples at concentrations well below the
TDWS and F&AL. Selenium was not detected in any samples.

o On the Clinch River, arsenic was detected in 58 of 62 samples at concentrations well below the TDWS
and F&AL. Thallium was detected in 2 of 62 samples at concentrations above the Tennessee Water
Quality Criteria (TWQC). Selenium was not detected in any sample.

o Onthe Emory River, arsenic was detected in 50 of 79 samples with all concentrations less than the
TDWS and the F&AL. Lead was detected in 49 of 80 samples with one result exceeding the TDWS.
Selenium was detected in three samples at concentrations less than the TDWS and F&AL. Thallium was
detected in 9 of 80 samples, with all results exceeding the TWQC.

e TVA and EPA continued the compilation of turbidity measurement data from the mouth of the Emory River to ERM
4.0. The data are being compiled and plotted on Mondays and Wednesdays to determine that turbidity curtain
placement is effective to prevent any adverse trends as the dredging operation continues.

e Plumes in the vicinity of dredging operations contain large amounts of suspended ash that settle out quickly and are
not representative of overall water quality in the Emory River. Comparing Plume results to those from Fixed Station
Locations provides an indication of the effectiveness of silt curtains and other Best Management Practices (BMPs)
employed by the project

¢ Swan Pond Embayment surface water sampling continues routinely twice per week at two locations along the Swan
Pond Embayment drainage ditches.

Sediment Sampling
e Vibecore sampling continued in support of dredging operations.

Air Sampling

o The majority of air monitoring results, including fixed-station and real-time monitoring results are below ambient air
quality criteria. Where instantaneous PM10 measurements above 150 micrograms per cubic meter (ug/ms) were
recorded, they were associated with wood fires or around railroad and utility work.

e TVA currently operates two real-time PM2.5 instruments: a tapered element oscillating microbalance (TEOM) PM2.5
instrument at Lakeshore Drive (PS07); and a PM2.5 Beta Attenuation Monitor (BAM) at the northern-most reach of
proposed work in the North Embayment (PS13). PM2.5 measurements at both stations for the past week averaged
below the action level of 26 ug/m®.

e Additional BAM instruments will be installed over the next few weeks at several other air monitoring stations.

e EPA s planning to collect co-located samples at two of the TVA air monitoring stations. Logistics are under review by
site EPA staff.
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Data Management

o Electronic data deliverables are flowing into the Equis database.

o Data validation continues for electronic data packages received from the laboratories.
e Migration of data from the TVA Equis to the EPA SCRIBE database is ongoing.

Biota Sampling

e Spring fish and benthic sampling was initiated the week of May 3" Additionally, larval fish are being collected by
Oak Ridge National Laboratory (ORNL) for a laboratory study.

e Monitoring of activity at bird boxes continues.

Non-Routine Sampling
e Used oil drums were sampled to support disposition.

FINAL REMEDIATION

e The Action Memorandum for the Non-Time-Critical Removal Action for the Embayment/Dredge Cell was finalized,
signed by TVA, and transmitted to EPA on Friday, April 14",

e The draft Removal Action Work Plan for the Non-Time-Critical Removal Action for the Embayment/Dredge Cell was
prepared for TVA review.

DREDGE CELL

e TVA, Jacobs, AECOM, TDEC, and EPA reviewing Stantec 90% Design Central Dredge Cell Working Platform
package.

EMBAYMENT ASH PROCESS

¢ Ash consolidation continued in the north embayment by TVA Civil Projects.

COMMUNICATIONS

Communications

e Posted to Website: Weekly Report and Weekly Area Resident Update, Public Meeting notice, and River Closure
Extension notice.

Published ad announcing public meeting.

Published ad announcing river closure extension.

Published ad listing April work plans.

Continued to work on addendum to the Community Involvement Plan with information about the Engineering
Evaluation/Cost Analysis.

Sent weekly update to members of the public who have signed up to receive it.

Continued to handle calls from the news media.

Finalized copy for next edition of the Dredge Report .

Finalized Kingston Ash Recovery Project update column for Roane County News.

Finalized plans for May 20" public meeting.

Outreach
¢ Provided updates on site activities to the Community Advisory Group.

e Updated Administrative Record with work plans and relevant documents.

e Continued to respond to calls and visits from residents.

¢ Updated message boards along Swan Pond Road announcing public meeting.

o Worked on Kingston Ash Recovery Project Site update column for Roane County News.
Other

o Distributed hard hat stickers to commemorate the shipment of 2 tons of ash.
e Worked on logo and slogan for celebration marking the end of time critical work.
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RIVER OPERATIONS

Weekly Average
Elevations and Flows

May 17, 2010
(7 day average of data through May 16)
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MEETINGS/CONFERENCE CALLS

Date Time Subject

05/17 0800 Environmental Group Meeting (ICP Conference Room)

05/17 0900 Risk Assessment (ICP Conference Room)

05/17 1000 Aquatic Resources Studies (Team Room)

05/17 1100 Leaching Test Protocols (Team Room)

05/17 1200 PMP (Team Room)

05/17 1200 MACTEC Safety Meeting (Trailer 1 Meeting Room)

05/17 1230 Public Meeting Q &A Remarks (ICP Conference Room)

05/17 1300 Environmental Staff Meeting (Team Room)

05/17 1400 Schedule Updates (ICP Conference Room)

05/18 0900 Discuss Presentation — Q & A for Public Meeting (ICP Conference Room)
05/18 0930 Communications (Team Room)

05/18 1000 TVA Communications (Team Room)

05/18 1330 Dredge Team Meeting (ICP Conference Room)

05/19 0900 Non-Time-Critical Weekly Meeting (ICP Conference Room)
05/19 0900 OIG Sevenson Audit (Team Room)

05/19 1100 MRC Environmental & Safety Meeting (ICP Conference Room)
05/19 1330 Test America Weekly Call (Conference Call)

05/19 1430 Data Management Meeting (ICP Conference Room)

05/20 1000 Pace Telephone (Conference Call)

05/20 1130 HSE Staff Meeting (ICP Conference Room)

05/20 1300 Jacobs OPR (ICP Conference Room)

05/20 1400 SOP Meeting (Team Room)

05/21 0900 Jacobs Senior Manager Meeting (ICP Conference Room)
Daily 0730 TVA Monitoring Wells Meeting (Trailer 1 Meeting Room)

Daily 0730 Site Safety Orientation (Old ICP Conference Room)

Daily 1500 Continuation of Dredge Progress & Contingency Plan (ICP Conference Room)
Mon-Thur 1600 Planning Meeting (ICP Conference Room)

Daily 1630 Ash Disposal Daily Recap (Team Room)
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Kingston Recovery Project

Date: May 10-16, 2010

Field Report

Safety Weekly Total Cumulative Total
Near Misses 0 47
First Aid Incidents 0 34
Recordable Incidents 0 10
Recordable Lost Time Incidents 0 12
Cenosphere/Debris Removal Weekly Total Cumulative Total
Light Debris Removal — Land
Outreach 0 18,867
(bags)
Divisions A, B,C,D Cleanup Division Shoreline Cleanup (bags) 1,140 104,192
Total Tons of Light Debris 0 959.57
Cenosphere Slurry Gallons 90,000 13,788,000
Environmental
Org Matrix Samples* Analyses* Results*
TVA Air - Real-Time -- -- 224,079
TVA Air — Fixed 2,139 -- 3,069
TVA Surface/Utility Water 3,680 32,019 154,897
TVA Ground Water (spring and well) 87 995 3,921
TVA Ash 78 104 1,689
TVA Soil/Sediment 252 416 3,032
TVA Biota 1,052 2,408 27,249
*Week ending May 14, 2010
Ash Dredging Weekly Cumulative
& Excavation Total Total
River E;g;”ated Cubic Yards (see 53,470 2,562,740
Ash Excavation |Estimated Cubic Yards
(east of Dike 2) |**Modified to an in-place volume 0 758,849
Ash Excavation |Estimated Cubic Yards ok
(other) **Modified to an in-place volume 0 73,000
Ash Disposal Estimated Tons 72,300 2,214,400

NOTE: Total dredged from the river has been adjusted to include ongoing survey adjustments and all of Sevenson (current
dredging contractor), TransAsh (prior dredging contractor), mechanical dredged from Sevenson and Aquarius, and TVA
reported hydraulic and mechanical dredging to date.
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